Author's response to reviews

Title: A draft framework for measuring progress towards the development of a national health information infrastructure

Authors:

Dean F. Sittig (dean.f.sittig@kp.org)
Richard N Shiffman (richard.shiffman@yale.edu)
Kevin Leonard (k.leonard@utoronto.ca)
Charles Friedman (friedmc1@mail.nlm.nih.gov)
Barbara Rudolph (brudolph@leapfroggroup.org)
George Hripcsak (hripcsak@columbia.edu)
Laura L Adams (ladams@riqi.org)
Lawrence C Kleinman (lkleinma@NEMOURS.ORG)
Rainu Kaushal (RKAUSHAL@PARTNERS.ORG)

Version: 5 Date: 31 May 2005

Author's response to reviews:

Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals).
done.

Headers- We are unable to accommodate headers in the full text version of the manuscript. Can we ask you to remove the headers from the manuscript?
done.

Manuscript sections - Manuscript sections should include (in the following order): Abstract; Background; Implementation; Results; Discussion (if applicable); Conclusions; Availability and requirements; Abbreviations (if any); Authors' contributions;

Have changed manuscript type to "debate". Have made the appropriate heading changes within the abstract and body of the paper.

Acknowledgements; References; Figure legends (if any); Tables (if any); Description of Additional files (if any).
done.

Conclusion- Please change the section heading to Conclusions.
see above now call Summary.

Availability and requirements- Pleas provide an 'Availability and requirements' section after the Conclusions.
see above. removed.

Competing interests -Please place this section after the Availability and requirements section.
done.

Figure legends- Please revise the figure legends provided in the manuscript text to ensure the title and legend accurately refer to the figure files.
done.

Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here: http://www.biomedcentral.com/info/ifora/cropping.
done.